



Article Neural Networks for the Detection of COVID-19 and Other Diseases: Prospects and Challenges

Muhammad Azeem¹, Shumaila Javaid², Ruhul Amin Khalil^{3,4}, Hamza Fahim², Turke Althobaiti⁵, Nasser Alsharif⁶ and Nasir Saeed^{4,*}

- School of Science, Engineering & Environment, University of Salford, Manchester M5 4WT, UK; azeemchaudharyg@gmail.com
- ² Department of Control Science and Engineering, College of Electronics and Information Engineering, Tongji University, Shanghai 201804, China; shumaila@tongji.edu.cn (S.J.); hamzafahim@tongji.edu.cn (H.F.)
- ³ Department of Electrical Engineering, University of Engineering and Technology, Peshawar 25120, Pakistan; ruhulamin@uetpeshawar.edu.pk
- ⁴ Department of Electrical and Communication Engineering, United Arab Emirates University (UAEU), Al-Ain 15551, United Arab Emirates
- ⁵ Department of Computer Science, Faculty of Science, Northern Border University, Arar 73222, Saudi Arabia; turke.althobaiti@nbu.edu.sa
- ⁶ Department of Administrative and Financial Sciences, Ranyah University Collage, Taif University, P.O. Box 11099, Taif 21944, Saudi Arabia; n.alsharif@tu.edu.sa
- * Correspondence: mr.nasir.saeed@ieee.org

Abstract: Artificial neural networks (ANNs) ability to learn, correct errors, and transform a large amount of raw data into beneficial medical decisions for treatment and care has increased in popularity for enhanced patient safety and quality of care. Therefore, this paper reviews the critical role of ANNs in providing valuable insights for patients' healthcare decisions and efficient disease diagnosis. We study different types of ANNs in the existing literature that advance ANNs' adaptation for complex applications. Specifically, we investigate ANNs' advances for predicting viral, cancer, skin, and COVID-19 diseases. Furthermore, we propose a deep convolutional neural network (CNN) model called ConXNet, based on chest radiography images, to improve the detection accuracy of COVID-19 disease. ConXNet is trained and tested using a chest radiography image dataset obtained from Kaggle, achieving more than 97% accuracy and 98% precision, which is better than other existing state-of-the-art models, such as DeTraC, U-Net, COVID MTNet, and COVID-Net, having 93.1%, 94.10%, 84.76%, and 90% accuracy and 94%, 95%, 85%, and 92% precision, respectively. The results show that the ConXNet model performed significantly well for a relatively large dataset compared with the aforementioned models. Moreover, the ConXNet model reduces the time complexity by using dropout layers and batch normalization techniques. Finally, we highlight future research directions and challenges, such as the complexity of the algorithms, insufficient available data, privacy and security, and integration of biosensing with ANNs. These research directions require considerable attention for improving the scope of ANNs for medical diagnostic and treatment applications.

Keywords: artificial neural networks; convolutional neural networks; healthcare; infectious diseases

1. Introduction

Artificial intelligence (AI) is revolutionizing the world with its endless applications. The platforms built on AI are prevalent, accelerating the pace of developing life-saving drugs and reducing operations costs. Artificial neural networks (ANN) are the building blocks of AI technologies, which simulate the human brain's analyzing and processing abilities to solve complex problems. The unique characteristics of ANN (such as efficient data handling, low complexity, reduced computation, and storage requirements) have enormous potential for a wide range of disciplines, including medical sciences [1] (especially in the areas of cardiology [2], radiology [3], oncology [4], urology [5]), veterinary [6],



Citation: Azeem, M.; Javaid, S.; Khalil, R.A.; Fahim, H.; Althobaiti, T.; Alsharif, N.; Saeed, N. Neural Networks for the Detection of COVID-19 and Other Diseases: Prospects and Challenges. *Bioengineering* **2023**, *10*, 850. https://doi.org/10.3390/ bioengineering10070850

Academic Editors: Victor Hugo C. De Albuquerque and Yunfeng Wu

Received: 29 May 2023 Revised: 9 July 2023 Accepted: 12 July 2023 Published: 18 July 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). stock exchange [7], law enforcement department [8], ecology [9], human resource management [10], signal processing with blind separation [11], and cybersecurity [12]. The ANNs are mainly based on mathematical models inspired by biological nervous systems, such as the brain's route information. They work similarly to an adaptive system, which updates its configuration in the learning phase and can be modelled for a specific application, such as data classification and pattern categorization. A neural network generally consists of three layers (i.e., input layer, hidden layer, and output layer). Based on these layers, ANN can be categorized into a single-layer feed-forward neural network (FFNN) [13], a multilayer feed-forward network [14], a single node with its feedback, and a recurrent multilayer network [15].

The massive potential of various types of ANN models in the healthcare domain must be addressed. Recent literature [16,17] has extensively studied the potential of ANN for the treatment and diagnosis of a wide range of infectious diseases, such as diarrhea [18], tuberculosis [19], dengue [20], COVID-19 [21,22], and childhood blindness [23]. State-ofthe-art schemes generally considered accuracy and complexity as the primary performance metrics to examine the performance of ANN.

The development of ANNs has revolutionized disease detection and diagnosis, offering numerous benefits over traditional approaches, having the capability to analyze large volumes of data, extract meaningful patterns, and provide accurate predictions. In the context of COVID-19, the rapid and precise identification of infected individuals is crucial for effective control and mitigation. In particular, those applied to radiography images can play a vital role in automating the analysis process and reducing the workload of healthcare professionals.

Therefore, in this paper, we have proposed a ConXNet model that can effectively classify COVID-19 cases from typical cases, providing timely and reliable support for diagnosis with the potential to transform healthcare practices, enabling efficient online monitoring and enhancing the overall quality of patient care. By highlighting the role of the ConXNet model in disease detection and emphasizing the specific application to COVID-19, our research contributes to the ongoing efforts in combating the pandemic and improving healthcare outcomes. The main contributions of our presented review are summarized as follows.

- First, we present a critical review of state-of-the-art ANN models that have contributed to the detection and diagnosis of various diseases, including skin diseases, retinal diseases, and COVID-19. While numerous studies are available on using ANNs in the medical field, a comprehensive review of these techniques is crucial to understand their current and future potential clearly. Therefore, this paper aims to offer a comprehensive and concise overview of recent advancements in ANNs for medical applications.
- Second, our work focuses explicitly on detecting and diagnosing COVID-19 using convolutional neural network (CNN) models. We provide an in-depth analysis of existing CNN models designed for COVID-19 detection, discussing their contributions and limitations.
- Then, we propose a novel deep learning model called ConXNet, which has been trained and tested using different datasets to improve the accuracy of COVID-19 detection by up to 98%. This contribution is significant as it offers an innovative approach to enhance the detection capabilities of AI-based models in the context of a global health crisis.
- Finally, we highlight the gaps that require attention in the future to improve ANNbased disease diagnosis and treatment. These future research challenges include algorithm complexity, inadequate available data, security and privacy concerns, and biosensing integration with ANNs.

The rest of the paper is organized as follows: Section 2 thoroughly discusses the literature background. The background of ANN is discussed in Section 3. Section 4 provides a comprehensive study of existing CNN models designed for COVID-19. Section 5 describes

our proposed ConXNet model, including the details of datasets and obtained results for COVID-19 detection. Section 6 reviews existing CNN models for various diseases. Section 7 provides the Discussion, Opportunities, and Open Issues; Section 8 highlights challenges and future research directions. Finally, Section 9 summarizes the findings of the presented reviews and provides future work directions.

2. Background and Related work

In recent years, numerous studies have focused on applying artificial intelligence (AI) techniques for combating the COVID-19 pandemic. In [24], the authors shed light on the opportunities, challenges, and prospects of explainable AI in fighting the pandemic. Their work emphasizes the potential of explainable AI techniques and how they can contribute to addressing the challenges posed by the pandemic. Similarly, [25] presents a scoping review of the challenges and opportunities of deep learning for cough-based COVID-19 diagnosis. By examining existing literature on deep learning approaches for diagnosing COVID-19 through cough analysis, they identify the challenges and discuss the potential of these techniques in improving diagnostic capabilities.

Researchers have explored various deep learning approaches for COVID-19 detection as AI evolves. In [26], the authors surveyed the detection of COVID-19 using deep learning techniques, providing insights into the different approaches employed and evaluating their cost-effectiveness. Furthermore, [27] presents a survey focusing on deep convolutional neural networks (CNNs) for detecting COVID-19 using medical images. Its review highlights the state-of-the-art deep learning models for COVID-19 detection through medical imaging, discussing their strengths, limitations, and potential applications in clinical practice.

For example, the authors in [28] investigated the effectiveness of ANN in detecting skin cancer. Their findings highlighted the exceptional performance of various neural network models, specifically the VGG-16 convolutional neural network (CNN). Notably, CNN achieved an accuracy of 87.6%. Moreover, the ability of advanced convolutional neural networks to distinguish between serious and benign skin cancers is examined in this study. In [29]; the authors performed an extensive study on neural networks for lung cancer detection and concluded that neural network techniques exhibit an excellent classification rate. However, they often require a significant amount of training time. In another work in [30], the authors studied a convolutional neural network, followed by a Pix2pix generative adversarial network (GAN) proposed for image enhancement, and the model successfully detected breast cancer with an accuracy of 78.52%. Although the limitations on the enormous volumes of annotated data may be reduced or even eliminated with the help of GANs, there are still some difficulties, such as the need for a more accurate feature extraction, which may have a significant impact on the practical implementation of ANN-based image processing techniques in digital pathology.

Furthermore, in [31], the authors focused on using convolutional neural networks (CNNs) for detecting malaria infection, a widely spread parasitic infectious disease. The importance of early detection in combating malaria was emphasized due to the limitations and errors associated with current techniques, such as manual microscopic examination and rapid diagnostic tests. Based on a custom CNN with three convolutional layers, the proposed model offers a faster and cost-effective method for distinguishing between healthy and infected blood samples, thus improving parasite (Plasmodium) detection. The proposed model achieved an accuracy of 96.71%. Similarly, in [32], the authors provided a comprehensive review of surveys and recent techniques in brain tumour classification, including preprocessing, feature extraction, and classification steps. The use of convolutional neural network models, transfer learning, and data augmentation, along with their achievements and limitations, is explored. The overview also highlights the importance of personalized and smart healthcare and suggests future research directions for improved brain tumor classification.

Moreover, in [33], the authors discussed various ML algorithms, such as naïve Bayes, classification and regression tree (CART), decision tree (DT), and support vector machine

(SVM), for different diseases, such as lung cancer, breast cancer, and skin diseases. In [34], the authors present a comparative study on data mining techniques for breast cancer prediction, including naïve Bayes, back-propagated neural networks, and decision tree algorithms. In [35], the authors investigated the effectiveness of numerous nature-inspired computing techniques, such as genetic algorithms [36], ant colony optimization [37], particle swarm optimization [38], and artificial bee colony [39], for diagnosing various critical human disorders. They concluded that nature-inspired computing techniques have high accuracy for disease detection and diagnosis. However, the survey lacks a detailed comparison with state-of-the-art schemes required to investigate the efficiency of nature-inspired techniques for real-world problems. Similarly, in [40], the authors extensively discuss hybrid models(i.e., neural networks combined with other classical methods or metaheuristic approaches) to outline selecting a well-suited ANN model for epidemic forecasts. Reference [41] shows that hybrid neural networks depict enhanced performance for epidemic forecasts.

Besides other infectious disease detections, recently, ANNs have been explored for COVID-19 detection. For instance, in [42], the authors highlight the role of AI, particularly in diagnosing and treating COVID-19. Moreover, it highlights that disease detection and diagnosis, virological research, drug and vaccine development, epidemic and transmission prediction, medical image analysis, and drug discovery are the primary areas that integrate AI to fight against COVID-19. Furthermore, the authors in [43] illustrated computer vision's role in combating the COVID-19 pandemic. They considered three types of visionary images for COVID-19 detection: computed tomography (CT) scans, X-ray imagery, and ultrasound imaging.

In addition to image-based diagnostics, researchers have investigated other modalities for COVID-19 detection. In [44], the authors propose a hybrid deep-fused learning approach to segregate infectious diseases, including COVID-19. Their study explores the combination of multiple data modalities to improve the accuracy of disease segregation. On the other hand, in [45], the authors introduce Epi-DNNs, which are deep neural networks informed by epidemiological priors. By integrating epidemiological priors into the neural network models, they aim to enhance the modelling of COVID-19 dynamics and improve forecasting capabilities.

Furthermore, the utilization of non-image-based approaches has also gained attention. In [46], the authors proposes a deep transfer learning-based CNN model for COVID-19 detection using computed tomography (CT) scan images. Their research focuses on improving the accuracy of COVID-19 detection through the application of transfer learning techniques in CT image analysis. Additionally, in [47], the authors investigated the diagnosis of COVID-19 from blood parameters using a CNN. Their study explores the potential of CNN models to analyze blood parameters and detect COVID-19, contributing to the development of non-image-based diagnostic approaches.

Our proposed work makes significant contributions in the context of this existing literature. We comprehensively review artificial neural network (ANN) models applied in medical domains, explicitly focusing on CNN models for COVID-19 detection. We propose a novel CNN model called ConXNet, which outperforms existing models in accuracy. Our work aims to bridge the gap between theory and practice, addressing the challenges and opportunities in COVID-19 detection. By incorporating insights from the literature and developing an innovative model, our research contributes to the ongoing efforts to combat the pandemic. It provides valuable insights for AI-driven disease detection and diagnosis.

3. Review of Artificial Neural Networks (ANNs)

Recently, neural networks have gained particular importance due to their diverse applications. Research and scientific communities are optimistic regarding the potential of ANN. The core property of an ANN is its capability of learning. There are three types of learning: (1) Supervised learning, which is accomplished in the existence of an observer. In this type of learning, a supervisor or observer is compulsory for error minimization.

(2) Unsupervised learning is accomplished without an observer's help, and the network itself discovers features, categories, patterns, or symmetries from the input data and associations for the input data over the output. (3) Reinforcement learning is used to train machine learning models. It enables an agent to learn through the consequences of actions in a specific environment.

Hence, many neural networks are developed and categorized over time based on their learning characteristics. In the following, we extensively discuss different types of neural networks to comprehend the background of ANN.

3.1. Perceptron

The perceptron model, proposed by Minsky and Papert in the 1950s, is one of the modest and ancient models of the neuron. Perceptron is the most minor component of a neural network that performs particular calculations to identify features in the input data [48]. The perceptron is a supervised learning algorithm that can easily classify the data into two categories by a hyperplane; therefore, it is also called a binary classifier. The perceptron algorithm has gained significant attention recently due to its use in establishing logic gates, such as AND, OR, or NAND.

Figure 1 shows a simple perceptron as an example with a layer of input nodes (layer 1) and a layer of output nodes (layer 2). Every perceptron layer is fully connected, but no links occur among nodes in the same layer. When layer 1 directs a signal to layer 2, the related weights on the links are applied, and each acceptance node on layer 2 sums up the received values. If the sum surpasses a given threshold, that node, in turn, directs an output signal.



Figure 1. Perceptron architecture.

The outputs are summed through all the inputs (a[i]) received by a node (j) in the output layer. The weighted sum is then combined with an activation function (f) to provide an output that can be either binary or continuous. Moreover, this weighted sum can be improved by including a bias value denoted as 'b.' Then, the output of each node is given by

$$S_{j} = f(\sum_{i=0}^{n} a_{i}w_{ij} + b).$$
(1)

in (1), if $S_j > \theta$, then $x_j = 1$; otherwise, $x_j = 0$, where θ is the threshold. The perceptron's output can be "trained" to match the desired output by adjusting the weights on the connections between layers. The amount of the correction is determined by multiplying the difference between the actual output (x[j]) and target (t[j]) values by a learning rate constant (C). If the nodes' output (a[i]) is 1, that connection weight is adjusted, and if it sends 0, it has no bearing on the output, and subsequently, there is no need for adjustment. Thus, the process weight adjustment is as follows:

where *C* is the learning rate. This training procedure is repeated until the network performance reaches a maximum threshold [49].

3.2. Multilayer Perceptron

In MLP, every node is associated with all neurons in the subsequent layer, making it a fully connected neural network. Input and output layers exist with many hidden layers (i.e., at least three or more layers) with bidirectional propagation (i.e., forward and backwards). In forward propagation, inputs are multiplied with weights and sent to the activation function, and in backpropagation, the weights are adjusted to reduce the loss [50]. Figure 2 shows a simple presentation of MLP, consisting of an input layer, a hidden layer, and an output layer. Except for the input nodes, each node is a neuron that uses a nonlinear activation function.





3.3. Feed-Forward Neural Network (FFNN)

FFNN [51] is the most basic neural network used for different applications, including computer vision, natural language processing, and speech recognition. In FFNN, data propagate only in one direction, passing through neural nodes and exiting through output nodes. FFNN comprises input and output layers; hidden layers may exist. The weights are static in FFNN, and an activation function is used by inputs multiplied by weights to learn the value of parameter θ . In addition, the design of FFNN requires considering several crucial factors, including an optimizer, cost function, and output units. FFNN in Figure 3 shows an input passed through one or more hidden layers unidirectionally, and weights are multiplied by inputs to minimize the error using an optimizer.



Figure 3. FFNN architecture.

3.4. Convolutional Neural Network (CNN)

Initially, CNN was only restricted to image processing, requiring large datasets. However, the success of AlexNet in the 2012 ImageNet challenge showed that the time has come to revisit CNNs, as large datasets available [52,53]. A CNN comprises a three-dimensional set of neurons instead of the typical two-dimensional array. The first layer is called a convolutional layer, where every neuron only routes the information from a minor part of the layer. The convolutional layer uses ReLU as an activation function, followed by softmax. The first layer is followed by a pooling layer, where the convolution layer's output drives to a fully connected neural network for classification. The CNNs show promising results in image and video recognition, semantic parsing, and paraphrase detection. Figure 4 visualizes a CNN model that first extracts all the features of the input image using the convolution operation and then uses the pooling layer to extract the most prominent features to pass through to the other convolutional layer for linearity. Later, the flattened layer shapes the input data for producing an output.



Figure 4. A simple presentation of CNN.

3.5. Radial Basis Function Neural Networks

The radial basis function (RBF) network comprises an input vector, followed by a layer of RBF neurons and an output layer with one node per class. Classification is accomplished by assessing the input's resemblance to data points from the training set, where each neuron stores a prototype, one of the samples from the training set. When a novel input vector (the n-dimensional vector that you are trying to classify) needs to be organised, every neuron computes the Euclidean distance between the input and its prototype [54,55]. Figure 5 shows the simple architecture of RBF comprising an input layer, a hidden layer (consisting of several RBF nonlinear activation units), and an output layer.



Figure 5. Architecture of RBF.

3.6. Recurrent Neural Networks (RNNs)

RNN is one of the most potent and robust state-of-the-art algorithms derived from FFNN for modelling sequence data [56]. RNN differs from other algorithms due to its internal memory and ability to process the sequence data lacking in different algorithms. Furthermore, RNN trains the model based on the current input and previous learning experience, offering more precise predictions.

Therefore, RNNs have promising applications in speech recognition, text summarization, prediction problems, face detection, music composition, and language processing [57]. Figure 6 shows a simple RNN architecture consisting of an input layer, one or more hidden layers, and an output layer.



Figure 6. Architecture of RNN.

3.6.1. Long Short-Term Memory (LSTM) Networks

LSTM is a type of RNN that comprises a "memory cell" that can preserve information in memory for long periods. A set of gates is used to regulate when information arrives in the memory of a cell. There are three types of gates, i.e., input gate, output gate, and forget gate. The input gate selects how much information from the last example will be reserved in memory; the output gate controls the amount of data delivered to the next layer, and forget gates regulate the tearing rate of memory keeping. This planning lets them learn longer-term reliance [58]. Figure 7 shows the architecture of LSTM consisting of three gates, i.e., forget gate, remember gate, and output gate, deciding the data sequence. In LSTM, the *tanh* activation function controls the values that flow across the network, whereas the sigmoid activation function (σ) is used at the gates as a gating function.



Figure 7. Architecture of LSTM.

3.6.2. Gated Recurrent Unit (GRU) Networks

The recurrent neural network (RNN) architecture, known as a "gated recurrent unit (GRU)", is intended to capture long-term dependencies in a sequential input efficiently. It was developed to improve the standard RNN model, solving issues including the vanishing gradient problem. Moreover, LSTM has three gates—input, output, and forget; GRU's bag only has two gates—reset and update. A GRU may learn to selectively update and preserve information over time by employing the update and reset gates, which enables it to capture long-term dependencies more efficiently than a conventional RNN [59]. GRUs are extensively used in several sequential data tasks, including machine translation, speech recognition, and natural language processing.

3.7. Sequence-to-Sequence Models

A sequence-to-sequence model comprises two RNNs with additional encoder and decoder modules. The encoder is used for the input data, and the decoder delivers the output. The encoder and decoder work in parallel using the same or different parameters [60]. Contrary to the actual RNN, this model only applies when the input data's length and output data's size are equal. These models are primarily used in chatbots, machine

translation, and question-answering systems, despite having similar help and curbs to the RNN. The layout of the sequence-to-sequence model, consisting of an encoder and a decoder, is shown in Figure 8. An encoder and a decoder are independent essential neural network models integrated into a massive network to create an output representation sequence. The encoder and decoder comprise two RNNs that act as an encoder and a decoder pair. The encoder receives an input of variable length and maps it to a fixed-length vector, then sends it to the decoder that maps the fixed-length vector back to a variable length as a target sequence to produce an output. This process is accomplished by utilizing the RNN network inside the encoder and decoder.



Figure 8. Architecture of sequence-to-sequence model.

3.8. Modular Neural Network

Another type of neural network is a modular neural network consisting of numerous dissimilar networks that can function self-sufficiently to complete various subtasks [61]. During the computation process, the different networks do not cooperate or signal each other to work independently to achieve the output. Thus, a large and complex computational procedure is performed faster by breaking it down into self-governing components. Modular neural networks are efficient and can conduct independent training [62]. Figure 9 shows the architecture of a modular neural network working on the principle of divide and conquer to split significant problems down into smaller manageable parts called modules to produce a single output for each using a module network. All the modules are trained independently, and then the production of each module is stored in a new NN model denoted as getting network in Figure 9 to produce an output.



Figure 9. Architecture of modular neural network.

4. CNN Models for COVID-19 Detection

In the previous section, we presented various neural networks. This section highlights the role of CNN in early predicting infectious diseases, especially COVID-19. Since the COVID-19 outbreak in China in December 2019, the global population and economy have been affected badly. To alleviate the spread of COVID-19, timely detection was essential to quarantine infected patients. The primary screening technique used to detect COVID-19 was polymerase chain reaction (PCR) testing; however, it is costly and can fall short of supply [63]. Therefore, an alternative screening method, such as a radiography examination, is used to detect COVID-19, which indicates that people affected with COVID-19 have irregularities in their chests, especially in the lungs. However, specialist radiologists are required to analyze such radiography images, and they might only sometimes be available. Therefore, there is an urgent need for an automated system that can analyze radiography images and lessen the workload of radiologists. In this case, neural-network-based X-ray screening is assumed to be a promising technique to test COVID-19 in asymptomatic

patients [64,65]. The leading interest in developing neural-network-based techniques for COVID-19 detection led to the development of many state-of-the-art neural network models to enhance COVID-19 detection accuracy [11,66–68]. Therefore, we provide a comprehensive review of existing neural network models to investigate the working of existing models in terms of their achievements and limitations provided in Table 1.

4.1. Decompose, Transfer, and Compose (DeTraC)

In [69], the authors present a deep CNN model (DeTraC) that indexes COVID-19 chest X-ray images. DeTraC can compact any abnormalities in the image dataset using a class decomposition mechanism. The DeTraC model consists of three phases. In the first phase, a backbone pre-trained CNN model of DeTraC is trained to fetch the deep local features of every image. In the second phase, a sophisticated gradient descent optimization process is used for training. In the last phase, the class composition layer is utilized to improve the final classification of images. The DeTraC model is trained and tested using a combination of two datasets (80 chest X-ray (CXR) images for regular patients and 116 for COVID-19 patients). The experimental results in [70] show that the DeTraC model exhibits a good accuracy of 93.1% (with a sensitivity of 100%) for identifying COVID-19 X-ray images from normal and severe acute respiratory syndrome cases. However, the small dataset (size) used for training and testing the model degraded its performance in complex scenarios.

4.2. COVID-Net

Recently, Want et al. used COVID-Net, a deep CNN model, to classify chest X-ray images into regular and COVID-19-infected patients [71]. To build COVID-Net, the humandriven principled network design prototyping is integrated with machine-driven design consideration to make a personalized network design for recognizing COVID-19 cases from chest radiography images. The authors of [72] used the COVIDx dataset, which comprises 13,975 chest radiography images, to train the proposed COVID-Net model. In [73], a built-in ResNet50 CNN model was first trained and tested on the COVIDx dataset, which achieved a test accuracy of 90.6%. Furthermore, the proposed COVID-Net model was then introduced and tested on the COVIDx dataset, which attained a test accuracy of 93.3%, indicating better performance than ResNet50 on the same dataset. The accuracy of COVID-Net can further be improved by different data augmentation techniques, utilizing network layers, and increasing the dataset size.

4.3. Coro-Net

The authors in [74] propose another deep CNN model, Coro-Net, to detect COVID-19 infection from chest X-ray images automatically. The foundation of Coro-Net is on Xception architecture, which is pre-trained on ImageNet and then trained on a combined dataset prepared using two publicly available datasets. The dataset used to train Coro-Net comprises 1300 images with an overall accuracy of 89.6%. Although Coro-Net shows promising results on a small dataset, its efficiency can be increased using a large dataset and minimum data preprocessing.

4.4. OptCoNet

The authors in [75] propose an optimized CNN for an automatic diagnosis of COVID-19 (OptCoNet). The OptCoNet building blocks have enhanced feature extraction and classification modules, and it uses a grey wolf optimizer (GWO) algorithm to optimize the hyperparameters to train the CNN layers. OptCoNet is trained using the datasets collected from two open-source repositories, comprising 2700 images with 900 COVID-19-infected patients' images and 1800 non-COVID-19 images. The results show that OptCoNet has promising results with an average accuracy of 97.78%. Nevertheless, the limited dataset used to train OptCoNet may lead to the inefficiency of the model.

4.5. COVID-MTNet

In [76], the authors present a deep learning neural network model called COVID-MTNet that uses multitask deep learning to detect COVID-19 from X-ray images. COVID-MTNet uses multiple models for different tasks, such as the classification model for COVID-19 detection and the segmentation model for region of interest (ROI) detection. Furthermore, the recurrent, residual neural network (RRCNN) model performs the COVID-19 detection task, and the NABLA-N network executes the infected region segmentation from X-ray and CT images. The dataset used to train the model comprises 5216 image samples. COVID-MTNet has shown 87.26% testing accuracy, which indicates good efficiency and reliability. However, data samples may be increased to validate the model's robustness and accuracy.

4.6. CovNet30

Another CNN-based model for automatically diagnosing COVID-19 from chest X-ray images is presented in [77]. During the training, several submodels are obtained from the Visual Geometry Group, composed of 19 layers, 16 convolutional layers, 3 fully connected layers, 5 MaxPooling layers, and 1 softmax layer to build a 30-layered CNN model (CovNet30), and the resulting submodels are arranged together using logistic regression. The CovNet30 model categorizes chest X-ray images into COVID-19, regular, and pneumonia classes and uses the COVID-19 CXR dataset to train the model. Their COVID-19 CXR dataset consists of 2764 chest X-ray images collected from three open-source data repositories, and it has achieved an accuracy of 92.74% for the classification of X-ray images.

4.7. COVIDPEN

In [78], the authors propose a COVID-19 detection technique called COVIDPEN, which uses chest X-rays and CT scans. COVIDPEN utilizes a transfer learning technique to identify COVID-19 patients. The dataset used to train the model comprises 746 image samples. COVIDPEN achieved an accuracy of 96% on the chest X-ray image dataset for the detection of COVID-19. Nevertheless, COVIDPEN also shows 96% accuracy only for a small dataset.

4.8. PDCOVIDNet

Another CNN model, Parallel-Dilated COVIDNet (PDCOVIDNet), is proposed in [79] for COVID-19 identification from chest X-ray images. PDCOVIDNet is trained on a dataset comprising 2905 chest X-ray images (COVID-19 cases and healthy controls). The proposed model achieved an average accuracy of 96.58%, indicating the high reliability of PDCOVID-Net. However, the dataset used to train and test the model is very small, and PDCOVIDNet will likely exhibit degradation in robustness and efficiency for larger datasets.

4.9. U-Net

In [80], a novel deep learning model was trained on chest X-ray images using U-Net architecture to detect COVID-19. The dataset used to evaluate the model comprises 1000 chest X-ray images comprising 552 normal images, while 448 are COVID-19 affected. Moreover, U-Net has achieved an overall accuracy of 94.10%. However, it shows poor performance for larger real-time datasets.

4.10. CapsNet

The authors in [81] present a novel ANN called CapsNet for identifying COVID-19 disease using chest X-ray images with capsule networks. Capsule networks preserve objects' spots and possessions in the image and model their relationships orderly to overcome the pooling layer issues of feature extraction (i.e., missing small features during feature passing to the next layer). Unfortunately, CapsNet also used a small dataset of 1050 images to train the model, and its performance may degrade for a larger dataset.

Ref.	Model	Dataset	Image Type	Accuracy	Comments
[66]	DeTraC deep convolutional neural network	80 negative 105 positive	Chest X-ray images	93.1%	This model produces a significant performance on chest radiography images. However, dataset samples are too small to make perception for its performance over a large dataset in an ideal condition.
[74]	Coro-Net deep neural network	310 negative 330 positive	Chest X-ray images	90%	The number of samples is not up to the mark to judge the model performance in a real-time environment. Although accuracy seems good, there are many chances of model overfitting in this case.
[75]	OptCoNet optimized convolutional neural network	1800 negative 900 positive	Chest X-ray images	97.78%	This model achieved significant accuracy. However, the dataset is average in size and cannot be considered ideal for complying the model for real-time analysis and set as an example.
[76]	COVID MTNet deep learning model	1341 negative 3875 positive	Chest X-ray and CT images	84.76%	The number of positive samples is relatively higher than the negative, , which may lead to an unusual behaviour model. Moreover, the model achieved an average accuracy, which may have high chances for underfitting of the model.
[79]	PDCOVIDNet parallel-dilated CNN	1341 negative 1564 Positive	Chest X-ray images	96.58%	The model achieved good accuracy on an average size of the dataset.
[80]	U-Net a novel deep learning model	552 negative 448 positive	Chest X-ray images	94.10%	This model acquires high accuracy on a small dataset, which needs to be better for a reliable model, and results may mislead in critical situations.
[81]	CapsNet a novel ANN model	1050 negative 231 positive	Chest X-ray images	97.24%	Again the model achieved higher accuracy on a small dataset. Consequently, model overfitting can be expected.
[82]	COVID-Net deep convolutional neural network	13,604 negative 2972 positive	Chest X-ray images	92.4%	This model seems quite promising with good accuracy results and a significant number of samples. However, a large set of negative samples can influence the results.
[83]	CovNet30 30-layered CNN model	1139 negative 1625 Positive	Chest X-ray images	92.74%	The model achieves good accuracy, but the number of samples is smaller compared with the performance of other state-of-the-art models trained on a large number of samples.
[84]	COVIDPEN pruned efficiently net-based model	180 negative 566 positive	Chest X-ray and CT images	96%	Due to the significantly small dataset size, despite achieving high accuracy, the reliability of the results is compromised.
[85]	EMCNet CNN and ensemble of machine learning classifiers	2300 negative 2300 positive	Chest X-ray images	98.91%	The model performed significantly well on a balanced dataset and produced high-accuracy results. However, increasing the data size may lead to a slight performance decrease.
[86]	Four-layered CNN model for analyzing CT images	3250 negative 5776 positive	Chest X-ray and CT images	97.8%	The model performed significantly well on a relatively large dataset. However, time and space complexity may vary over time, and several training hyperparameters are also high, which may cause the slow performance of a model in the future with more large datasets.
	ConXNet proposed CNN model	10,192 negative 3616 Positive	Chest X-ray images	97.8%	A proposed model in the paper acquires a significantly good accuracy on a relatively large dataset as compared with other models. Some of the models also achieved good results close to the proposed model. However, this model is trained on a fewer number of hyperparameters by using different techniques, such as batch processing and dropout layers, to overcome overfitting of the model as well. Therefore, the results of the model are reliable but can be improved in the future with more samples.

 Table 1. CNN models for COVID-19 detection.

5. Proposed ConXNet Model for COVID-19 Detection

In the previous section, we examined various CNN models for COVID-19 detection. We introduce our new ConXNet model, which offers enhanced accuracy and precision. This model has been trained on diverse datasets of chest X-ray images. The following section presents this novel ConXNet model for COVID-19 detection.

5.1. Dataset

As mentioned earlier, we have used several datasets to evaluate the performance of the ConXNet model. The dataset comprises 13,808 chest X-ray images with 3616 COVID-19 data and 10,192 average or non-COVID-19 data [87]. Furthermore, to ensure a robust evaluation of our proposed model, we conducted testing using diverse and representative datasets obtained from publicly available sources. These datasets encompass a wide range of authentic clinical scenarios to reflect the challenges encountered in medical imaging. Specifically, we gathered a comprehensive collection of 2473 chest X-ray (CXR) images from the podcast dataset, which provides a rich and diverse set of patient cases. In addition, we incorporated 183 CXR images obtained from a renowned German medical school, further enhancing the applicability of our evaluation. To further expand the scope and diversity of our assessment, we incorporated 559 CXR images sourced from Kaggle, a popular platform for data science competitions. These images were selected to encompass various pathologies and variations encountered in routine clinical practice. Lastly, we augmented our dataset with an additional 400 CXR images obtained from a reliable source on GitHub. This inclusion allows us to address a more comprehensive array of clinical scenarios and strengthen the validity of the evaluation. By leveraging these diverse and authentic datasets [88–94], we aimed to thoroughly assess the performance and generalizability of our proposed model, ensuring its reliability and applicability.

5.2. ConXNet Architecture

The ConXNet model consists of four blocks, where each block comprises a convolutional layer (Conv), rectified linear unit (ReLU) (operating as an activation function), batch normalization, and MaxPooling layer. Figure 10 represents a detailed architecture of the proposed ConXNet model. X-ray images are fed to the first Conv layer as an input to extract the features (edges, soft edges, blur) from the given input. Once the features are removed, the Conv layer produces a filter matrix called a feature map as an output. After applying different filters on the input image, ReLU is used for nonlinear operations so that the model can learn non-negative linear values. Later on, the rectified feature map is passed through the MaxPooling layer to fetch the most prominent element. Furthermore, batch normalization techniques are applied to prevent the model from overfitting. Once all the convolutional operations are performed, the output is flattened before being sent to the fully connected dense layer to produce the final output. The last, output layer classifies COVID-19 or normal X-ray images from our data.



Figure 10. Proposed ConXNet model for COVID-19 detection.

5.3. Experimental Results

We have implemented the CNN model in a 64-bit Windows 10 operating system using Python 3.6. The tensor flow framework builds and trains the proposed model using Keras as the back end. The flowchart provided in Figure 11 deliberately explains the overall flow diagram of our proposed model. Before fitting the model, similar images from both classes are selected, such as 3500 for COVID images and 3500 for typical images. A total of 7000 images are used (according to the machine compatibility) for training and evaluation of the proposed model. Furthermore, the dataset is split into 70% (4900) images for training and 30% (2100) images for testing purposes.

Additionally, the model is compiled using binary cross-entropy (BCE) as a loss function for binary classification. Epochs of 100 and batch sizes 32 are used to fit the model, while 0.001 is used as a preliminary learning rate value. Finally, the Adam optimizer reduces the error rate and automatically tunes the learning rate. It is an efficient variant of gradient descent that prevents the model from hand-tuning the learning rate and does it by itself more quickly and efficiently. The results show that this CNN model achieves an overall accuracy of 97.8%, significantly promising for large datasets. Moreover, we have also evaluated the performance of the proposed model by carrying out a real-world evaluation test. In the following, we discuss the results of our test.



Figure 11. Flowchart of the proposed algorithm.

5.3.1. Obtained Results for "COVID-19_Radiography_Dataset"

To effectively observe the performance of the proposed model, the dataset is divided into training and testing with a ratio of 70% and 30%, respectively. The Tensorflow framework builds and trains the proposed model using Keras as the back end. The model is compiled using binary cross-entropy and accuracy performance metrics. Furthermore, learning is first initialized to 0.001 preliminaries to train the model, while 100 epochs are used with 32 batch sizes. Moreover, the error rate is optimized using the Adam optimizer for authentic learning. Finally, three more common performance measures (accuracy, precision, and F1-score) are used to observe the proposed model's efficiency. It is observed that the model achieved a significant accuracy score of 97.8% with precision and an F1-score of 97.93% and 97.92%, respectively. Table 2 summarizes the results produced by the proposed ConXNet model.

Table 2. Performance measures of ConXNet model.

Epochs	Accuracy	Precision	F1-Measure
100	97.8%	97.93%	97.92%

5.3.2. Testing of the Proposed Scheme

After training and testing the proposed model on COVID-19_Radiography_Dataset, we also tested the proposed scheme to evaluate the efficiency and accuracy of the model on unseen data selected from the test dataset. The real-world evaluation experiment gives random images as input to classify whether the images belong to COVID-positive or negative patients. The chest X-ray image is passed to the network as input, and then the image is preprocessed to match the target size input for the proposed model. Moreover, the proposed model predicts the probabilities of the given image, whether it belongs to the COVID class or normal class, with an approximate predicted percentage. The heat map effect is used to visualize the area in the image affected by the disease. These experiments give us a valuable view to evaluate the efficiency of the proposed model in the real world. Figure 12a Illustrates an original X-ray image of a COVID-19 patient, and Figure 12b shows the results after detecting images with heat map highlights of infected regions. Meanwhile, Figure 12c illustrates the original image of the typical patient, and Figure 12d shows after evaluation of the results. To facilitate the reproducibility of our work, and we have made the complete code and trained model available online (https://github.com/azeemchaudhary/ CovidXNet/blob/main/Covid_Detection_Model.ipynb (20 May 2023)).



Figure 12. Images used to evaluate the model accuracy with their outputs (**a**). Original image of COVID-19 patient (**b**). Heat map view of COVID-infected region. (**c**) Original image of the normal patient (**d**). Heat map view of the normal image.

6. CNN Models for Detection of Other Diseases

CNN models play a crucial role in Alzheimer's, cancer, and retinal diseases by analyzing medical images and extracting patterns for accurate diagnosis and treatment. In Alzheimer's, CNN models analyze brain imaging data to identify abnormalities, track disease progression, and predict outcomes. For cancer, CNN models aid in tumor detection, classification, and staging. In retinal diseases, CNN models diagnose conditions such as diabetic retinopathy and age-related macular degeneration by analyzing retinal images. With their ability to learn complex patterns, CNN models significantly contribute to improving medical imaging analysis, enabling informed decision-making by healthcare professionals and enhancing patient outcomes. This section briefly discusses the role of CNN models in these diseases to highlight their global impact, diagnostic challenges, the complexity of medical images, and research significance, as accurate diagnosis is crucial for these diseases, which profoundly impact health.

6.1. Alzheimer's Disease Detection

Several studies have investigated using CNN to improve Alzheimer's disease detection and diagnosis. For instance, in [95], the authors propose an automated and reliable deep learning model for detecting and identifying Alzheimer's disease using magnetic resonance imaging (MRI) and positron emission tomography (PET) that achieved 94.48% average accuracy. In another work [96], an ensemble deep CNN model was developed to detect Alzheimer's disease using the Open Access Series of Imaging Studies (OASIS) dataset that acquired 93.18% accuracy. In [97], a CNN model developed for Alzheimer's disease detection uses a brain imaging structured dataset comprising brain MRI images that achieved 96% accuracy, indicating excellent efficiency compared with other state-of-the-art models. In [98], a four-way CNN-based classifier was used that attains 98.8% accuracy by classifying four classes, including Alzheimer's disease, mild cognitive impairment, late mild cognitive impairment, and no disease using the ADNI dataset. Recently, Alzheimer's disease has been detected by introducing transfer learning from a dataset of 2D images to 3D CNNs to enable early diagnosis of Alzheimer's disease using MRI imaging datasets [99].

6.2. Cancer Detection

Deep learning methods, such as CNN, can extract hierarchical features from image data without manual selection and are successfully applied in cancer tissue detection. For example, early breast cancer detection, diagnosis, and treatment are possible due to a computer-aided diagnostic (CAD) system based on mammograms analyzed by CNN-based classifiers. In [100], a CAD approach based on in-depth features detected breast cancer using a mammogram image dataset, achieving 81.75% accuracy. In [101], a novel CNN-based model detected lung cancer using computed tomography (CT) scan images and achieved 86.6% accuracy. In [102], a novel CNN-based model is proposed for detecting prostate cancer using the diffusion-weighted magnetic resonance imaging (DWI) dataset, which acquired 84% accuracy. In [103], a fully convolutional network (FCN) was introduced to detect liver cancer using CT scan images, achieving 86% accuracy using a threefold cross-validation technique.

Besides breast cancer detection, CNN has also shown great potential for brain tumor detection, the most frequent and severe type of cancer with a life expectancy of only a few months in the most advanced stages. Therefore, treatment planning for brain tumors is essential in improving patients' quality of life. In [104], a CNN classification model is proposed for automatic brain tumor detection using brain MRI image datasets. The experimental results show that the proposed model acquired 97.5% accuracy. Another CNN model presented in [105] classifies images for brain tumor detection and achieves 97.87% accuracy. The CNN model presented in [106] performs the classification and segmentation of brain tumors using the MRI image dataset and achieves 94.6% accuracy. In [107], a CNN-based model extracted brain cancers from MRI images with an overall accuracy of 96%. A CNN model for detecting benign tumors was recently introduced in [108] that was trained and tested using BraTS2013 and WBA datasets and delivered 96–99% accuracy.

6.3. Retinal Disease Detection

Optical coherence tomography (OCT) imaging makes accurate and reliable diagnoses of retinal disorders, which is critical for clinical relevance. In [109], the authors introduced a novel CNN model for accurate detection and classification into normal, drusen macular degeneration, and diabetic macular edema using an optical coherence tomography (OCT) imaging dataset. Moreover, the Kuan filter is used to input OCT pictures first to reduce intrinsic speckle noise. Furthermore, hyperparameter optimization techniques are used to optimize the CNN network, and K-fold validation is performed to ensure complete utilization of the dataset. As a result, the presented model achieved 95.7% accuracy. Another CNN model presented in [110] accurately detects retinal blood vessels. Recently, in [111], the CNN-based approach localizes, identifies, and quantifies abnormal features in the eye retina using the OCT image dataset and achieves an accuracy of 95.8%. In [112], OCT imaging of the retinas was analyzed using three different CNN models comprising five, seven, and nine layers to identify the various retinal layers, extract usable information, detect new aberrations, and predict several eye abnormalities. Furthermore, the CNN model recently developed in [113] classifies retinal disease using OCT images and achieves 98.73% accuracy. Table 3 summarizes the contributions of the CNN models for other diseases' detection.

Table 3. Summary of the contributions of CNN models for other diseases' detection.

Diseases	Ref.	Model	Dataset	Accuracy
Alzheimer	[114]	Deep CNN for Alzheimer's disease detection	Total 615 MRI scan images	94.48%
	[115]	An ensemble of deep CNN	OASIS dataset comprises 416 MRI scan images	93.18%
	[116]	CNN for classification of Alzheimer's disease	ADNI dataset comprises a total of 1455 MRI scan images	96%
	[117]	A Deep CNN based multiclass classification of Alzheimer's disease	Dataset comprises of total 355 MRI scans images	98.8%
Cancer	[118]	Breast cancer detection using extreme learning machine based on feature fusion with CNN deep features	Total 400 mammography images	81.75%
	[119]	Lung cancer detection and class- ification with 3D CNN	Kaggle Data Science Bowl (DSB) comprises a total of 1397 MRI scan images	86.6%
	[120]	Prostate cancer detection using deep CNN	Diffusion-weighted magnetic resonance imaging dataset comprises a total of 427 images	84%
	[121]	Fully CNN for liver segmentation and lesions detection	Dataset comprises of total 88 CT scan images	86%
	[122]	Brain tumor classification using CNN	Radiopaedia and brain tumor image segmentation benchmark 2015 datasets are used	97.5%
	[123]	Brain tumor detection using CNN	BRATS dataset comprises 217 MRI images	97.87%
	[124]	Brain tumor classification and segmentation using faster R-CNN	A total of 218 MRI images are used	94.6%
	[125]	Brain tumor classification in magnetic resonance images using deep learning and wavelet transform	MRI image dataset	96%
	[126]	Deriving tumor detection models using CNN from MRI of human brain scans	BraTS2013 dataset and WBA dataset are used	96-99%
Retinal	[127]	Deep CNN framework for retinal disease diagnosis	OCT image dataset containing a total of 12,000 images	95.7%
	[111]	Detection of retinal abnormalities using CNN	Dataset comprises a total of 1110 fundus images	95.8%
	[112]	DL-CNN-based approach diagnosis of retinal diseases	OCT image dataset comprises a total of 84,495 images	96.5%
	[113]	Retinal disease classification using CNN algorithm	OCT image dataset comprises a total of 108,312 images	98.73%

7. Discussion, Opportunities, and Open Issues

7.1. Discussion

Applying ANNs and deep learning in COVID-19 detection has shown remarkable potential in improving diagnostics and patient care. These models have demonstrated high accuracy and speed in analyzing medical images, allowing for rapid identification of COVID-19 cases. However, it is crucial to acknowledge the limitations and challenges associated with these techniques. The interpretability of deep learning models remains a significant concern, as their decision-making process is often considered a "black box". Efforts should be focused on developing explainable AI approaches that provide insights into the features and patterns driving the model's predictions. This would enhance the trust and acceptance of AI-based systems in clinical practice.

Another critical aspect is the ethical use of AI in healthcare. Privacy concerns, data security, and potential biases in AI algorithms must be carefully addressed. Healthcare professionals and AI experts should collaborate to develop guidelines and regulations that ensure patient privacy, informed consent, and appropriate deployment of AI technologies. Transparency and accountability in AI decision-making processes are essential for building public trust and ensuring equitable access to healthcare services. It is also crucial to mitigate biases in training data that could result in COVID-19 and other disease detection and treatment disparities.

Collaboration between interdisciplinary teams is vital to advancing the field of AI in COVID-19 detection. Close cooperation between clinicians, researchers, data scientists, and policymakers is necessary to leverage the potential of AI in a responsible and effective manner. Combining clinical expertise with AI capabilities, we can develop models that improve diagnostic accuracy and provide valuable clinical insights and decision support. Multidisciplinary collaborations can also help address challenges, such as data sharing, standardization of protocols, and validation of AI models across diverse healthcare settings.

7.2. Opportunities

The presented review and the proposed model in this paper can serve as a guideline and framework for designing and implementing AI-based COVID-19 and other disease detection systems. It provides a comprehensive overview of the various phases of developing such systems and the key considerations and specifications for each phase. By following the reference model, researchers and developers can ensure a systematic and structured approach to the design and implementation process.

A critical aspect of the reference model is the inclusion of performance measures. These measures provide a quantitative assessment of the effectiveness and reliability of AI-based COVID-19 detection systems. Examples of performance measures may include sensitivity, specificity, accuracy, precision, recall, and F1-score. By defining and evaluating these measures, researchers can assess the performance of their models and compare them with existing approaches, enabling a fair and objective evaluation.

7.2.1. Integration of AI with Internet of Healthcare Things (IoHT)

Integrating AI algorithms with the IoHT, such as wearable sensors and remote monitoring systems, presents real-time data collection and analysis opportunities. By leveraging IoT data; AI models can continuously monitor COVID-19 patients, enabling early detection of deterioration and timely intervention. This integration also opens avenues for personalized medicine and remote patient management, enhancing healthcare accessibility and reducing the burden on healthcare systems. Integrating COVID detection technologies with current technologies, such as the Internet of Healthcare Things (IoHT), presents numerous research opportunities and challenges. One aspect to consider is the tradeoff between learning cost and performance. Deep learning, while achieving good predictive scores, can be computationally expensive. This raises questions about whether opting for weak learners with good features or complex learners with raw data is more beneficial. A potential approach could involve leveraging learned features offline and using weak learners online for real-time decision-making.

Additionally, latency is critical in disease detection systems, as timely data processing is essential for accurate and efficient diagnosis. This calls for investigations into optimizing the tradeoff between learning cost and latency. Techniques such as edge computing can be explored, where data preprocessing and initial feature extraction are performed at the edge devices, reducing the need for transmitting large amounts of data to the cloud. This can help mitigate latency issues and improve real-time decision-making.

In the context of communications, the issue of latency and data processing delays becomes even more relevant. Integrating COVID detection systems with communication networks requires careful consideration of network architecture, protocols, and transmission capabilities. Techniques such as edge caching and distributed processing can be explored to minimize latency and improve the overall system performance.

7.2.2. Federated Learning for Privacy-Preserving Analysis

Federated learning allows collaborative model training on distributed data sources while preserving privacy. This approach can be beneficial in COVID-19, where data privacy is crucial. Researchers can leverage data from multiple institutions without compromising patient privacy by developing federated learning frameworks. This collective analysis can lead to improved models with enhanced generalizability and robustness.

7.2.3. Integration of AI in Telemedicine Platforms

Telemedicine has witnessed significant growth during the COVID-19 pandemic. Integrating AI capabilities into telemedicine platforms can enhance remote diagnostics and triaging. AI models can assist healthcare providers in making accurate and timely decisions based on patient data and medical images. Combining telemedicine and AI can extend healthcare services to remote areas, improving access to quality care and reducing healthcare disparities.

7.3. Open Issues

Research, collaboration, and ethical considerations are needed in AI-based COVID-19 detection. Addressing these issues will improve the accuracy and interpretability of AI models and ensure their responsible integration into healthcare systems. By focusing on standardization, interpretability, ethics, and practical integration, we can unlock the full potential of AI in combating the COVID-19 pandemic and future healthcare challenges.

7.3.1. Standardization of Evaluation Metrics and Benchmark Datasets

To enable fair comparisons and promote reproducibility, it is crucial to establish standardized evaluation metrics and benchmark datasets for COVID-19 detection models. Consensus on evaluation protocols, such as sensitivity, specificity, and area under the receiver operating characteristic curve (AUC-ROC), will facilitate meaningful comparisons of different AI models. Additionally, developing diverse and representative benchmark datasets will ensure the generalizability of models across other populations and imaging modalities.

7.3.2. Addressing the Challenges of Model Interpretability and Explainability

Enhancing the interpretability of AI models is vital for gaining the trust of healthcare professionals and end users. Methods for interpreting and explaining deep learning models' decisions, such as attention mechanisms, saliency mapping, and rule-based explanations, must be further explored. By understanding the reasoning behind AI predictions, clinicians can make more informed decisions and identify potential limitations or biases in the models. Research efforts should focus on developing interpretable AI models that can provide transparent and understandable insights.

7.3.3. Ethical Considerations in AI deployment

The ethical implications of AI in healthcare, including privacy, bias, and transparency, must be addressed. Robust guidelines and regulations should be developed to ensure the responsible use of AI technologies in COVID-19 detection. Ethical frameworks should prioritize patient privacy, informed consent, and equitable access to healthcare resources. Additionally, efforts should be made to address potential biases in training data that could result in disparities in COVID-19 diagnosis and treatment. Discussions and collaborations

between healthcare professionals, policymakers, and AI researchers are essential to develop ethical guidelines that align with societal values.

7.3.4. Integration of AI Models into Clinical Workflows

Successful integration of AI models into clinical practice requires addressing practical challenges. User-friendly interfaces, seamless integration with existing healthcare systems, and validation of AI models in diverse clinical settings are crucial steps. Collaboration between AI researchers and healthcare providers is essential to understand the clinical needs and adapt AI technologies accordingly. Moreover, regulatory considerations and compliance with healthcare standards must be considered to ensure AI's safe and effective use in real-world healthcare environments. Ongoing research and collaboration are needed to overcome these challenges and enable the practical deployment of AI models in clinical workflows.

8. Future Research Directions

In this section, we discuss future research directions and challenges, including algorithms complexity, inadequate available data, security and privacy issues, and biosensing integration with ANNs that require considerable attention for enhancing the scope of ANNs for medical treatment and diagnosis.

8.1. Complexity

The complexity of neural-network-based disease detection and diagnosis models depends on the computational time and number of samples used to train the network. Since, in medical imagining applications, the accurate detection of disease is more critical, the analysis complexity (such as time and computational complexities) is compromised in most cases [128]. However, in the future, neural-network-based disease detection applications can introduce different methods to address the complex issue of preventing excessive resource consumption without degrading accuracy. For example, 1D and 2D filters can be used at the feature-extraction image-restoration layer with high-resolution images to reduce the complexity [129]. Furthermore, data transformation techniques, including data augmentation [130], variable standardization [131], and proper initialization of ANN [132], can also play a significant role in reducing complexity since these techniques use limited data samples to achieve better results without compromising on accuracy. Moreover, limiting the number of neurons during model training [133], dropout layer technique [134], network pruning techniques [135], normalization techniques, transfer learning techniques, and brute-force exploration approach [136] can also be used to minimize complexity.

8.2. Algorithm Selection

Another challenging issue for neural-network-based infectious disease detection is the selection of the proper algorithm. ANN algorithms are generally categorized as supervised learning and unsupervised learning, compromising of various algorithms, including decision trees (DT) [137], naïve Bayes (NB) [138], support vector machine (SVM) [139], random forest (RF) [140], k-nearest neighbour (KNN) [139], and k-means [141] for executing different operations. Moreover, algorithm selection based on the dataset also plays an essential role in achieving the desired results.

8.3. Deficient Training Data

In addition to algorithm selection, another major issue is obtaining valuable information from the raw data. The ANNs are primarily applied to large datasets and composite models that need extensive training to obtain the required training effects [142]. However, acquiring adequate training data is challenging due to large data unavailability in numerous domains, such as disease detection using image classification. Thus, the need for more available data for training and evaluating the parameters in the neural networks results in network inefficiency and overfitting. For instance, blood pressure measurements in most of the COVID-19 critical patients are erroneous and unstable, which limits the amount of accurate data.

8.4. Privacy and Security

One of the key issues with using ANNs for infectious disease detection is privacy and security concerns. Nevertheless, [143] introduced several methods to attain privacy from the "trusted" authority, attackers, and involved entities. Recently, for COVID-19 location-aware apps, a polling mechanism is used to ensure the privacy of the patient infected with COVID-19 from the noninfected patient [144]. This allows noninfected patients to poll the health authority frequently to check whether they have been in close contact with infected patients. Moreover, privacy in the healthcare domain can also be achieved using private messaging systems [145], private set intersection protocols [146], and memo verification methods using cryptography [147]. However, these protocols are computationally intensive, requiring a tradeoff between security and computational efficiency [148]. Therefore, in the future low computational techniques, such as blockchain [149], cloud computing [150], and deep belief neural networks (DBNN), [151] requires more focus to ensure the required level of privacy and security.

8.5. Integration of Biosensing and ANN

The tremendous progress in nanotechnology over the past few years has also advanced the development of biosensors for medical applications [152,153]. Biosensors are analytical tools that can observe cellular activity at the nanoscale and identify specific biological strains at the cellular level [22,154]. To overcome the difficulties of intelligent information identification in a biological medium for contagious disease detection, research communities are concentrating on combining AI cognition techniques, such as metaheuristic algorithms and ANN models with biosensing applications [155,156]. Moreover, ANN-integrated biosensing applications can also considerably increase the accuracy and reliability of disease diagnosis as biosensors can generate time-series data that can be trained and tested employing cutting-edge NNs [157].

9. Conclusions

This paper thoroughly reviewed the advances in infectious disease detection and diagnostics by ANNs. The detailed discussions reveal that ANN's ability to solve complex problems has a huge potential to improve disease detection and diagnostic accuracy, leading to advanced patient care and treatment. Besides reviewing the existing ANN-based disease detection models, we propose a new model, ConXNet, which considerably enhances the detection accuracy of COVID-19 patients with low hyperparameters and lesser time complexity. The extensive testing and training of ConXNet using different available datasets show that ConXNet can detect COVID-19 patients significantly well. Finally, we presented the challenges linked to disease prediction using neural networks and future directions. Because this work is mainly focused on the ANN-based disease detection models, it will be interesting to investigate the potential of ANN for targeted drug delivery and advanced surgical procedure in the future. Moreover, we also suggest testing different evolving techniques, such as transfer learning and hybrid schemes, to enhance detection accuracy and precision in medical diagnosis.

Author Contributions: Conceptualization, N.S.; data curation, S.J., R.A.K. and H.F.; formal analysis, T.A. and N.S.; funding acquisition, N.A.; investigation, M.A.; methodology, M.A.; project administration, H.F., T.A. and N.A.; resources, N.A.; supervision, N.S.; validation, S.J. and R.A.K.; writing—original draft, M.A.; writing—review & editing, S.J., R.A.K., H.F., T.A. and N.A. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are available on request from the corresponding author.

Conflicts of Interest: The authors declare no conflicts of interest.

Abbreviations

The following abbreviations are used in this manuscript:

1D	one (1) dimensional
2D	two (2) dimensional
3D	three (3) dimensional
AI	artificial intelligence
ANN	artificial neural network
CART	classification and regression tree
CNN	convolutional neural network
CT	computed tomography
DBNN	deep belief neural networks
DeTraC	Decompose, Transfer, and Compose
DT	decision tree
FNN	feed-forward neural network
GRU	gated recurrent unit
KNN	k-nearest neighbor
LSTM	long short-term memory
ML	machine learning
MLP	multilayer perceptron
NB	naïve Bayes
NN	neural network
PCR	polymerase chain reaction
PNN	probabilistic neural networks
RBF	radial basis function
ReLU	rectified linear unit
RF	random forest
RRCNN	recurrent, residual neural network
ROI	region of interest
RNN	recurrent neural network
SVM	support vector machine

References

- Baloni, D.; Verma, S.K. Detection of hydrocephalus using deep convolutional neural network in medical science. *Multimed. Tools Appl.* 2022, *81*, 16171–16193. [CrossRef]
- 2. Itchhaporia, D. Artificial intelligence in cardiology. Trends Cardiovasc. Med. 2022, 32, 34–41. [CrossRef] [PubMed]
- 3. Hasan, Z.; Key, S.; Habib, A.R.; Wong, E.; Aweidah, L.; Kumar, A.; Sacks, R.; Singh, N. Convolutional neural networks in ENT Radiology: Systematic review of the literature. *Ann. Otol. Rhinol. Laryngol.* **2023**, *132*, 417–430. [CrossRef] [PubMed]
- Mehranian, A.; Wollenweber, S.D.; Walker, M.D.; Bradley, K.M.; Fielding, P.A.; Su, K.H.; Johnsen, R.; Kotasidis, F.; Jansen, F.P.; McGowan, D.R. Image enhancement of whole-body oncology [18F]-FDG PET scans using deep neural networks to reduce noise. *Eur. J. Nucl. Med. Mol. Imaging* 2022, 49, 539–549. [CrossRef] [PubMed]
- Ostrowski, D.A.; Logan, J.R.; Antony, M.; Broms, R.; Weiss, D.A.; Van Batavia, J.; Long, C.J.; Smith, A.L.; Zderic, S.A.; Edwins, R.C.; et al. Automated Society of Fetal Urology (SFU) grading of hydronephrosis on ultrasound imaging using a convolutional neural network. *J. Pediatr. Urol.* 2023, *in press.* [CrossRef]
- Pereira, A.I.; Franco-Gonçalo, P.; Leite, P.; Ribeiro, A.; Alves-Pimenta, M.S.; Colaço, B.; Loureiro, C.; Gonçalves, L.; Filipe, V.; Ginja, M. Artificial Intelligence in Veterinary Imaging: An Overview. *Vet. Sci.* 2023, *10*, 320. [CrossRef]
- Kurani, A.; Doshi, P.; Vakharia, A.; Shah, M. A comprehensive comparative study of artificial neural network (ANN) and support vector machines (SVM) on stock forecasting. *Ann. Data Sci.* 2023, 10, 183–208. [CrossRef]
- Kosan, M.A.; Karacan, H.; Urgen, B.A. Predicting personality traits with semantic structures and LSTM-based neural networks. *Alex. Eng. J.* 2022, *61*, 8007–8025. [CrossRef]
- 9. Wei, W.; Zhang, Q. Evaluation of rural financial, ecological environment based on machine learning and improved neural networks. *Neural Comput. Appl.* **2022**, *34*, 9335–9352. [CrossRef]

- 10. Zhang, N. Benefit evaluation of human resource management in agricultural enterprises based on convolutional neural network. *Pak. J. Agric. Sci.* **2023**, *60*, 217–227.
- 11. Khan, J.; Jan, T.; Khalil, R.; Altalbe, A. Hybrid source prior based independent vector analysis for blind separation of speech signals. *IEEE Access.* **2020**, *8*, 132871–132881. [CrossRef]
- 12. He, H.; Ji, Y.; Huang, H.H. Illuminati: Towards explaining graph neural networks for cybersecurity analysis. In Proceedings of the 2022 IEEE 7th European Symposium on Security and Privacy (EuroS&P), Genoa, Italy, 6–10 June 2022; pp. 74–89.
- Ananya, S.; Bharamagoudra, M.R.; Bharath, K.; Pujari, R.R.; Hanamanal, V.A. Glaucoma Detection using HOG and Feed-forward Neural Network. In Proceedings of the 2023 IEEE International Conference on Integrated Circuits and Communication Systems (ICICACS), Raichur, India, 24–25 February 2023; pp. 1–5.
- 14. Pilarz, J.; Polishuk, I.; Chorążewski, M. Prediction of sound velocity for selected ionic liquids using a multilayer feed-forward neural network. *J. Mol. Liq.* **2022**, *347*, 118376. [CrossRef]
- 15. Liu, Z.; Tahir, G.A.; Masuyama, N.; Kakudi, H.A.; Fu, Z.; Pasupa, K. Error-output recurrent multi-layer Kernel Reservoir Network for electricity load time series forecasting. *Eng. Appl. Artif. Intell.* **2023**, *117*, 105611. [CrossRef]
- Cheng, K.; Li, Z.; He, Y.; Guo, Q.; Lu, Y.; Gu, S.; Wu, H. Potential use of artificial intelligence in infectious disease: Take ChatGPT as an example. *Ann. Biomed. Eng.* 2023, *51*, 1130–1135. [CrossRef]
- 17. Mishra, S.; Kumar, R.; Tiwari, S.K.; Ranjan, P. Machine learning approaches in the diagnosis of infectious diseases: A review. *Bull. Electr. Eng. Inform.* **2022**, *11*, 3509–3520. [CrossRef]
- Ghaffari, M.; Monneret, A.; Hammon, H.; Post, C.; Müller, U.; Frieten, D.; Gerbert, C.; Dusel, G.; Koch, C. Deep convolutional neural networks for the detection of diarrhea and respiratory disease in preweaning dairy calves using data from automated milk feeders. J. Dairy Sci. 2022, 105, 9882–9895. [CrossRef]
- 19. An, L.; Peng, K.; Yang, X.; Huang, P.; Luo, Y.; Feng, P.; Wei, B. E-TBNet: Light Deep Neural Network for automatic detection of tuberculosis with X-ray DR Imaging. *Sensors* **2022**, *22*, 821. [CrossRef]
- 20. Panja, M.; Chakraborty, T.; Nadim, S.S.; Ghosh, I.; Kumar, U.; Liu, N. An ensemble neural network approach to forecast Dengue outbreak based on climatic condition. *Chaos Solitons Fractals* **2023**, *167*, 113124. [CrossRef]
- Yang, L.; Wang, S.H.; Zhang, Y.D. EDNC: Ensemble deep neural network for COVID-19 recognition. *Tomography* 2022, *8*, 869–890. [CrossRef]
- 22. Bhatti, D.M.S.; Khalil, R.A.; Saeed, N.; Nam, H. Detection and Spatial Correlation Analysis of infectious Diseases using wireless body area network under Imperfect wireless channel. *Big Data* **2022**, *10*, 54–64. [CrossRef]
- 23. Shen, Y.; Luo, Z.; Xu, M.; Liang, Z.; Fan, X.; Lu, X. Automated detection for Retinopathy of Prematurity with knowledge distilling from multi-stream fusion network. *Knowl.-Based Syst.* **2023**, *269*, 110461. [CrossRef]
- Abiodun, K.M.; Awotunde, J.B.; Aremu, D.R.; Adeniyi, E.A. Explainable AI for fighting COVID-19 pandemic: Opportunities, challenges, and future prospects. In *Computational Intelligence for COVID-19 and Future Pandemics: Emerging Applications and Strategies*; Springer: Singapore, 2022; pp. 315–332.
- Ghrabli, S.; Elgendi, M.; Menon, C. Challenges and opportunities of deep learning for cough-based COVID-19 diagnosis: A scoping review. *Diagnostics* 2022, 12, 2142. [CrossRef] [PubMed]
- MV, M.K.; Atalla, S.; Almuraqab, N.; Moonesar, I.A. Detection of COVID-19 using deep learning techniques and cost-effectiveness evaluation: A survey. *Front. Artif. Intell.* 2022, 5, 912022.
- Khattab, R.; Abdelmaksoud, I.R.; Abdelrazek, S. Deep Convolutional Neural Networks for Detecting COVID-19 Using Medical Images: A Survey. New Gener. Comput. 2023, 41, 343–400. [CrossRef]
- Malo, D.C.; Rahman, M.M.; Mahbub, J.; Khan, M.M. Skin Cancer Detection using Convolutional Neural Network. In Proceedings of the 2022 IEEE 12th Annual Computing and Communication Workshop and Conference (CCWC), Las Vegas, NV, USA, 26–29 January 2022; pp. 169–176.
- Thaseen, M.; UmaMaheswaran, S.; Naik, D.A.; Aware, M.S.; Pundhir, P.; Pant, B. A Review of Using CNN Approach for Lung Cancer Detection Through Machine Learning. In Proceedings of the 2022 2nd International Conference on Advance Computing and Innovative Technologies in Engineering (ICACITE), Greater Noida, India, 28–29 April 2022; pp. 1236–1239. [CrossRef]
- Aizaz, Z.; Khare, K.; Khursheed, A.; Tirmizi, A. Pix2Pix Generative adversarial Networks (GAN) for breast cancer detection. In Proceedings of the 2022 5th International Conference on Multimedia, Signal Processing and Communication Technologies (IMPACT), Aligarh, India, 26–27 November 2022; pp. 1–5. [CrossRef]
- Ahmed, K.M.T.; Rahman, Z.; Shaikh, R.; Hossain, S.I. Malaria Parasite Detection Using CNN-Based Ensemble Technique on Blood Smear Images. In Proceedings of the 2023 International Conference on Electrical, Computer and Communication Engineering (ECCE), Chittagong, Bangladesh, 23–25 February 2023; pp. 1–4. [CrossRef]
- Singh, Y.P.; Lobiyal, D.K. Brain Tumor Classification Using Deep Transfer Learning CNN Models. In Proceedings of the 2022 4th International Conference on Inventive Research in Computing Applications (ICIRCA), Coimbatore, India, 21–23 September 2022; pp. 891–897. [CrossRef]
- 33. Kumar, Y.; Koul, A.; Singla, R.; Ijaz, M.F. Artificial intelligence in disease diagnosis: A systematic literature review, synthesizing framework and future research agenda. *J. Ambient. Intell. Humaniz. Comput.* **2022**, *14*, 8459–8486. [CrossRef]
- Mishra, A.; Khan, M.; Khan, W.; Khan, M.Z.; Srivastava, N.K. A Comparative Study on Data Mining Approach Using Machine Learning Techniques: Prediction Perspective. In *Pervasive Healthcare: A Compendium of Critical Factors for Success*; Springer: Cham, Switzerland, 2022; pp. 153–165.

- 35. Kumar, S.; Singh, H. Antenna Array Pattern Synthesis Using Nature-Inspired Computational Techniques: A Review. *Arch. Comput. Methods Eng.* **2023**, *30*, 3235–3269. [CrossRef]
- Sharma, S.; Kumar, V. Application of genetic algorithms in healthcare: A review. In Next Generation Healthcare Informatics; Springer: Singapore, 2022; pp. 75–86.
- Qi, A.; Zhao, D.; Yu, F.; Heidari, A.A.; Wu, Z.; Cai, Z.; Alenezi, F.; Mansour, R.F.; Chen, H.; Chen, M. Directional mutation and crossover boosted ant colony optimization with application to COVID-19 X-ray image segmentation. *Comput. Biol. Med.* 2022, 148, 105810. [CrossRef]
- Shami, T.M.; El-Saleh, A.A.; Alswaitti, M.; Al-Tashi, Q.; Summakieh, M.A.; Mirjalili, S. Particle swarm optimization: A comprehensive survey. *IEEE Access* 2022, 10, 10031–10061. [CrossRef]
- 39. Kaya, E.; Gorkemli, B.; Akay, B.; Karaboga, D. A review on the studies employing artificial bee colony algorithm to solve combinatorial optimization problems. *Eng. Appl. Artif. Intell.* **2022**, *115*, 105311. [CrossRef]
- 40. Mukhtar, M.; Oluwasanmi, A.; Yimen, N.; Qinxiu, Z.; Ukwuoma, C.C.; Ezurike, B.; Bamisile, O. Development and comparison of two novel hybrid neural network models for hourly solar radiation prediction. *Appl. Sci.* **2022**, *12*, 1435. [CrossRef]
- Li, C.; Changsheng, Z.; Liu, J.; Rong, Y. Application of neural-network hybrid models in estimating the infection functions of nonlinear epidemic models. *arXiv* 2022, arXiv:2203.05018.
- 42. Floresta, G.; Zagni, C.; Gentile, D.; Patamia, V.; Rescifina, A. Artificial intelligence technologies for COVID-19 de novo drug design. *Int. J. Mol. Sci.* 2022, 23, 3261. [CrossRef]
- Li, H.; Zeng, N.; Wu, P.; Clawson, K. Cov-Net: A computer-aided diagnosis method for recognizing COVID-19 from chest X-ray images via machine vision. *Expert Syst. Appl.* 2022, 207, 118029. [CrossRef]
- 44. Rasheed, J.; Alsubai, S. A Hybrid Deep Fused Learning Approach to Segregate Infectious Diseases. *Comput. Mater. Contin.* 2023, 74, 4239–4259. [CrossRef]
- 45. Ning, X.; Jia, L.; Wei, Y.; Li, X.A.; Chen, F. Epi-DNNs: Epidemiological priors informed deep neural networks for modeling COVID-19 dynamics. *Comput. Biol. Med.* **2023**, *158*, 106693. [CrossRef]
- 46. Kathamuthu, N.D.; Subramaniam, S.; Le, Q.H.; Muthusamy, S.; Panchal, H.; Sundararajan, S.C.M.; Alrubaie, A.J.; Zahra, M.M.A. A deep transfer learning-based convolution neural network model for COVID-19 detection using computed tomography scan images for medical applications. *Adv. Eng. Softw.* 2023, 175, 103317. [CrossRef]
- Erol Doğan, G.; Uzbaş, B. Diagnosis of COVID-19 from blood parameters using convolutional neural network. *Soft Comput.* 2023, 27, 10555–10570. [CrossRef]
- 48. Park, Y.R.; Murray, T.J.; Chen, C. Predicting sun spots using a layered perceptron neural network. *IEEE Trans. Neural Netw.* **1996**, 7, 501–505. [CrossRef]
- Devi, M.S.; Goni, S.; Tasneem, P.; Vintha, K.; Kumar, D.S. Unipolar and Bipolar Mathematical Inference of Weight Adjustment Mode of Single Layer Perceptron on AND Logic Gate. In Proceedings of the 2023 Third International Conference on Advances in Electrical, Computing, Communication and Sustainable Technologies (ICAECT), Bhilai, India, 5–6 January 2023; pp. 1–5.
- Sivasankari, S.; Surendiran, J.; Yuvaraj, N.; Ramkumar, M.; Ravi, C.; Vidhya, R. Classification of diabetes using multilayer perceptron. In Proceedings of the 2022 IEEE International Conference on Distributed Computing and Electrical Circuits and Electronics (ICDCECE), Ballari, India, 23–24 April 2022; pp. 1–5.
- Xue, Y.; Tong, Y.; Neri, F. An ensemble of differential evolution and Adam for training feed-forward neural networks. *Inf. Sci.* 2022, 608, 453–471. [CrossRef]
- 52. Zhang, X.; Zhou, J.; Sun, W.; Jha, S.K. A Lightweight CNN Based on Transfer Learning for COVID-19 Diagnosis. *Comput. Mater. Contin.* **2022**, 72, 1123–1137. [CrossRef]
- 53. Khalil, R.; Jones, E.; Babar, M.; Jan T.; Zafar, M.; Alhussain, T. Speech emotion recognition using deep learning techniques: A review. *IEEE Access.* 2019, 7, 117327–117345. [CrossRef]
- 54. Heidari, A.; Navimipour, N.J.; Unal, M. A Secure Intrusion Detection Platform Using Blockchain and Radial Basis Function Neural Networks for Internet of Drones. *IEEE Internet Things J.* **2023**, *10*, 8445–8454. [CrossRef]
- 55. Olabanjo, O.A.; Wusu, A.S.; Manuel, M. A machine learning prediction of academic performance of secondary school students using radial basis function neural network. *Trends Neurosci. Educ.* **2022**, *29*, 100190. [CrossRef] [PubMed]
- Liu, X.; Wang, Y.; Chen, X.; Gales, M.J.; Woodland, P.C. Efficient lattice rescoring using recurrent neural network language models. In Proceedings of the 2014 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), Florence, Italy, 4–9 May 2014; pp. 4908–4912.
- 57. Zhu, J.; Jiang, Q.; Shen, Y.; Qian, C.; Xu, F.; Zhu, Q. Application of recurrent neural network to mechanical fault diagnosis: A review. J. Mech. Sci. Technol. 2022, 36, 527–542. [CrossRef]
- Huang, R.; Wei, C.; Wang, B.; Yang, J.; Xu, X.; Wu, S.; Huang, S. Well performance prediction based on Long Short-Term Memory (LSTM) neural network. J. Pet. Sci. Eng. 2022, 208, 109686. [CrossRef]
- Zhang, W.; Li, H.; Tang, L.; Gu, X.; Wang, L.; Wang, L. Displacement prediction of Jiuxianping landslide using gated recurrent unit (GRU) networks. Acta Geotech. 2022, 17, 1367–1382. [CrossRef]
- 60. Gong, S.; Li, M.; Feng, J.; Wu, Z.; Kong, L. Diffuseq: Sequence to sequence text generation with diffusion models. *arXiv* 2022, arXiv:2210.08933.
- 61. Varela-Santos, S.; Melin, P. A new modular neural network approach with fuzzy response integration for lung disease classification based on multiple objective feature optimization in chest X-ray images. *Expert Syst. Appl.* **2021**, *168*, 114361. [CrossRef]

- 62. Dahouda, M.K.; Joe, I. Neural architecture search net-based feature extraction with modular neural network for image classification of copper/cobalt raw minerals. *IEEE Access* 2022, 10, 72253–72262. [CrossRef]
- Banerjee, A.; Sarkar, A.; Roy, S.; Singh, P.K.; Sarkar, R. COVID-19 chest X-ray detection through blending ensemble of CNN snapshots. *Biomed. Signal Process. Control.* 2022, 78, 104000. [CrossRef]
- 64. Kollias, D.; Arsenos, A.; Kollias, S. Ai-mia: COVID-19 detection and severity analysis through medical imaging. In *European Conference on Computer Vision*; Springer: Cham, Switzerland, 2022; pp. 677–690.
- 65. Kogilavani, S.V.; Prabhu, J.; Sandhiya, R.; Kumar, M.S.; Subramaniam, U.; Karthick, A.; Muhibbullah, M.; Imam, S.B.S. COVID-19 detection based on lung CT scan using deep learning techniques. *Comput. Math. Methods Med.* 2022, 2022, 7672196. [CrossRef]
- 66. Abbas, A.; Abdelsamea, M.M.; Gaber, M.M. Classification of COVID-19 in chest X-ray images using DeTraC deep convolutional neural network. *Appl. Intell.* **2021**, *51*, 854–864. [CrossRef]
- 67. Wieczorek, M.; Siłka, J.; Woźniak, M. Neural network powered COVID-19 spread forecasting model. *Chaos Solitons Fractals* **2020**, 140, 110203. [CrossRef]
- Videla, L.S.; Harita, U.; Chippada, N.; Santhi, C.; Gupta, A.S.G. Convolution Neural Networks based COVID-19 Detection using X-ray Images of Human Chest. In Proceedings of the 2022 8th International Conference on Smart Structures and Systems (ICSSS), Chennai, India, 21–22 April 2022; pp. 1–4.
- Jia, F.; Mao, S.; Huang, Z.; Zeng, T. Detachable image decomposition and illumination mapping search for low-light image enhancement. J. Comput. Appl. Math. 2023, 436, 115435. [CrossRef]
- Al-Itbi, A.S.; Alwahhab, A.B.A.; Sahan, A.M. X-ray Covid-19 Detection Based on Scatter Wavelet Transform and Dense Deep Neural Network. *Comput. Syst. Sci. Eng.* 2022, 41, 1255–1271. [CrossRef]
- Marks, K.J.; Whitaker, M.; Anglin, O.; Milucky, J.; Patel, K.; Pham, H.; Chai, S.J.; Kirley, P.D.; Armistead, I.; McLafferty, S.; et al. Hospitalizations of children and adolescents with laboratory-confirmed COVID-19—COVID-NET, 14 states, July 2021–January 2022. Morb. Mortal. Wkly. Rep. 2022, 71, 271. [CrossRef]
- 72. Yang, X.; He, X.; Zhao, J.; Zhang, Y.; Zhang, S.; Xie, P. COVID-CT-dataset: A CT scan dataset about COVID-19. *arXiv* 2020, arXiv:2003.13865.
- Theckedath, D.; Sedamkar, R. Detecting affect states using VGG16, ResNet50 and SE-ResNet50 networks. SN Comput. Sci. 2020, 1, 79. [CrossRef]
- Khan, A.I.; Shah, J.L.; Bhat, M.M. CoroNet: A deep neural network for detection and diagnosis of COVID-19 from chest X-ray images. Comput. Methods Programs Biomed. 2020, 196, 105581. [CrossRef]
- Goel, T.; Murugan, R.; Mirjalili, S.; Chakrabartty, D.K. OptCoNet: An optimized convolutional neural network for an automatic diagnosis of COVID-19. *Appl. Intell.* 2021, *51*, 1351–1366. [CrossRef]
- Alom, M.Z.; Rahman, M.; Nasrin, M.S.; Taha, T.M.; Asari, V.K. COVID_MTNet: COVID-19 detection with multi-task deep learning approaches. *arXiv* 2020, arXiv:2004.03747.
- Agrawal, R.; Sarkar, H.; Prasad, A.O.; Sahoo, A.K.; Vidyarthi, A.; Barik, R.K. Exploration of Deep Neural Networks and Effect of Optimizer for Pulmonary Disease Diagnosis. SN Comput. Sci. 2023, 4, 471. [CrossRef]
- Lv, J.; Zhu, F.; Lu, K.; Wang, W.; Zhang, J.; Chen, P.; Zhao, Y.; Wu, Z. COVID-19 Classification from Chest X-rays Based on Attention and Knowledge Distillation. In *International Conference on Intelligent Computing*; Springer: Cham, Switzerland, 2022; pp. 787–798.
- Chowdhury, N.K.; Rahman, M.M.; Kabir, M.A. PDCOVIDNet: A parallel-dilated convolutional neural network architecture for detecting COVID-19 from chest X-ray images. *Health Inf. Sci. Syst.* 2020, *8*, 27. [CrossRef] [PubMed]
- Kalane, P.; Patil, S.; Patil, B.; Sharma, D.P. Automatic detection of COVID-19 disease using U-Net architecture based fully convolutional network. *Biomed. Signal Process. Control.* 2021, 67, 102518. [CrossRef]
- Toraman, S.; Alakus, T.B.; Turkoglu, I. Convolutional capsnet: A novel artificial neural network approach to detect COVID-19 disease from X-ray images using capsule networks. *Chaos Solitons Fractals* 2020, 140, 110122. [CrossRef]
- Wang, L.; Lin, Z.Q.; Wong, A. Covid-net: A tailored deep convolutional neural network design for detection of COVID-19 cases from chest X-ray images. *Sci. Rep.* 2020, 10, 19549. [CrossRef]
- 83. Gour, M.; Jain, S. Stacked convolutional neural network for diagnosis of COVID-19 disease from X-ray images. *arXiv* 2020, arXiv:2006.13817.
- 84. Jaiswal, A.K.; Tiwari, P.; Rathi, V.K.; Qian, J.; Pandey, H.M.; Albuquerque, V.H.C. Covidpen: A novel covid-19 detection model using chest x-rays and ct scans. *Medrxiv* 2020. [CrossRef]
- 85. Saha, P.; Sadi, M.S.; Islam, M.M. EMCNet: Automated COVID-19 diagnosis from X-ray images using convolutional neural network and ensemble of machine learning classifiers. *Inform. Med. Unlocked* **2021**, 22, 100505. [CrossRef]
- Azeem, M.; Javaid, S.; Fahim, H.; Saeed, N. AI for Good: CNN for COVID-19 Detection, Diagnosis, and Personalized Treatment. *IEEE Smart Cities Newsletter*. 2021. Available online: https://smartcities.ieee.org/newsletter/july-2021/ai-for-good-cnn-for-covid-19-detection-diagnosis-and-personalized-treatment (accessed on: 28 June 2023).
- 87. Kaggle. COVID-19 Radiography Database; Kaggle: San Francisco, CA, USA, 2020.
- BIMCV-COVID19. Available online: https://bimcv.cipf.es/bimcv-projects/bimcv-covid19/#1590858128006-9e640421-6711 (accessed on 29 December 2021).
- COVID-19-Image. Available online: https://github.com/ml-workgroup/covid-19-image-repository/tree/master/png (accessed on 1 January 2021).

- 90. COVID-19 DATABASE-SIRM. Available online: https://sirm.org/category/senza-categoria/covid-19/ (accessed on 20 December 2021).
- 91. Covi19 Image. Available online: https://eurorad.org (accessed on 1 January 2021).
- 92. Covid Chest X-ray Dataset. Available online: https://github.com/ieee8023/covid-chestxray-dataset (accessed on 20 December 2021).
- 93. COVID-19 Chest X-ray Image. 2021. Available online: https://figshare.com/articles/COVID-19_Chest_X-Ray_Image_ Repository/12580328 (accessed on 20 December 2021).
- 94. COVID-CXNet. Available online: https://github.com/armiro/COVID-CXNet (accessed on 20 December 2021).
- 95. Chui, K.T.; Gupta, B.B.; Alhalabi, W.; Alzahrani, F.S. An MRI scans-based Alzheimer's disease detection via convolutional neural network and transfer learning. *Diagnostics* **2022**, *12*, 1531. [CrossRef]
- 96. Patil, V.; Madgi, M.; Kiran, A. Early prediction of Alzheimer's disease using convolutional neural network: A review. *Egypt. J. Neurol. Psychiatry Neurosurg.* **2022**, *58*, 130. [CrossRef]
- 97. Liu, Z.; Lu, H.; Pan, X.; Xu, M.; Lan, R.; Luo, X. Diagnosis of Alzheimer's disease via an attention-based multi-scale convolutional neural network. *Knowl.-Based Syst.* 2022, 238, 107942. [CrossRef]
- Houria, L.; Belkhamsa, N.; Cherfa, A.; Cherfa, Y. Multi-modality MRI for Alzheimer's disease detection using deep learning. *Phys. Eng. Sci. Med.* 2022, 45, 1043–1053. [CrossRef]
- 99. Ebrahimi, A.; Luo, S.; Alzheimer's Disease Neuroimaging Initiative. Convolutional neural networks for Alzheimer's disease detection on MRI images. J. Med. Imaging 2021, 8, 024503. [CrossRef]
- Popescu, D.; El-Khatib, M.; El-Khatib, H.; Ichim, L. New trends in melanoma detection using neural networks: A systematic review. *Sensors* 2022, 22, 496. [CrossRef]
- Gunjan, V.K.; Singh, N.; Shaik, F.; Roy, S. Detection of lung cancer in CT scans using grey wolf optimization algorithm and recurrent neural network. *Health Technol.* 2022, 12, 1197–1210. [CrossRef]
- 102. Lamrani, D.; Cherradi, B.; El Gannour, O.; Bouqentar, M.A.; Bahatti, L. Brain tumor detection using mri images and convolutional neural network. *Int. J. Adv. Comput. Sci. Appl.* **2022**, *13*, 452–460. [CrossRef]
- Shahin, A.I.; Aly, W.; Aly, S. MBTFCN: A novel modular fully convolutional network for MRI brain tumor multi-classification. *Expert Syst. Appl.* 2023, 212, 118776. [CrossRef]
- Akinyelu, A.A.; Zaccagna, F.; Grist, J.T.; Castelli, M.; Rundo, L. Brain tumor diagnosis using machine learning, convolutional neural networks, capsule neural networks and vision transformers, applied to MRI: A survey. J. Imaging 2022, 8, 205. [CrossRef]
- 105. Tiwari, P.; Pant, B.; Elarabawy, M.M.; Abd-Elnaby, M.; Mohd, N.; Dhiman, G.; Sharma, S. Cnn based multiclass brain tumor detection using medical imaging. *Comput. Intell. Neurosci.* 2022, 2022, 1830010. [CrossRef]
- 106. Vankdothu, R.; Hameed, M.A.; Fatima, H. A brain tumor identification and classification using deep learning based on CNN-LSTM method. *Comput. Electr. Eng.* 2022, 101, 107960. [CrossRef]
- Chattopadhyay, A.; Maitra, M. MRI-based brain tumour image detection using CNN based deep learning method. *Neurosci. Inform.* 2022, 2, 100060. [CrossRef]
- Saeedi, S.; Rezayi, S.; Keshavarz, H.; R Niakan Kalhori, S. MRI-based brain tumor detection using convolutional deep learning methods and chosen machine learning techniques. *BMC Med. Inform. Decis. Mak.* 2023, 23, 16. [CrossRef] [PubMed]
- 109. Puneet; Kumar, R.; Gupta, M. Optical coherence tomography image-based eye disease detection using deep convolutional neural network. *Health Inf. Sci. Syst.* 2022, *10*, 13. [CrossRef]
- 110. Sarki, R.; Ahmed, K.; Wang, H.; Zhang, Y.; Wang, K. Convolutional neural network for multi-class classification of diabetic eye disease. *EAI Endorsed Trans. Scalable Inf. Syst.* 2022, 9, e5. [CrossRef]
- 111. Akil, M.; Elloumi, Y.; Kachouri, R. Detection of retinal abnormalities in fundus image using CNN deep learning networks. In *State of the Art in Neural Networks and Their Applications*; Elsevier: Amsterdam, The Netherlands, 2021; pp. 19–61.
- 112. Tayal, A.; Gupta, J.; Solanki, A.; Bisht, K.; Nayyar, A.; Masud, M. DL-CNN-based approach with image processing techniques for diagnosis of retinal diseases. *Multimed. Syst.* 2022, 28, 1417–1438. [CrossRef]
- 113. Mittal, P.; Bhatnaghar, C. Retinal Disease Classification Using Convolutional Neural Networks Algorithm. *Turk. J. Comput. Math. Educ. (TURCOMAT)* **2021**, *12*, 5681–5689.
- Vu, T.D.; Ho, N.H.; Yang, H.J.; Kim, J.; Song, H.C. Non-white matter tissue extraction and deep convolutional neural network for Alzheimer's disease detection. *Soft Comput.* 2018, 22, 6825–6833. [CrossRef]
- 115. Islam, J.; Zhang, Y. An ensemble of deep convolutional neural networks for Alzheimer's disease detection and classification. *arXiv* **2017**, arXiv:1712.01675.
- 116. Wen, J.; Thibeau-Sutre, E.; Diaz-Melo, M.; Samper-González, J.; Routier, A.; Bottani, S.; Dormont, D.; Durrleman, S.; Burgos, N.; Colliot, O.; et al. Convolutional neural networks for classification of Alzheimer's disease: Overview and reproducible evaluation. *Med. Image Anal.* 2020, *63*, 101694. [CrossRef]
- 117. Farooq, A.; Anwar, S.; Awais, M.; Rehman, S. A deep CNN based multi-class classification of Alzheimer's disease using MRI. In Proceedings of the 2017 IEEE International Conference on Imaging systems and techniques (IST), Beijing, China, 18–20 October 2017; pp. 1–6.
- 118. Wang, Z.; Li, M.; Wang, H.; Jiang, H.; Yao, Y.; Zhang, H.; Xin, J. Breast cancer detection using extreme learning machine based on feature fusion with CNN deep features. *IEEE Access* 2019, 7, 105146–105158. [CrossRef]
- 119. Alakwaa, W.; Nassef, M.; Badr, A. Lung cancer detection and classification with 3D convolutional neural network (3D-CNN). *Lung Cancer* 2017, *8*, 409. [CrossRef]

- 120. Yoo, S.; Gujrathi, I.; Haider, M.A.; Khalvati, F. Prostate cancer detection using deep convolutional neural networks. *Sci. Rep.* **2019**, *9*, 19518. [CrossRef]
- 121. Ben-Cohen, A.; Diamant, I.; Klang, E.; Amitai, M.; Greenspan, H. Fully convolutional network for liver segmentation and lesions detection. In *Deep Learning and Data Labeling for Medical Applications*; Springer: Cham, Switzerland, 2016; pp. 77–85.
- 122. Abiwinanda, N.; Hanif, M.; Hesaputra, S.T.; Handayani, A.; Mengko, T.R. Brain tumor classification using convolutional neural network. In World Congress on Medical Physics and Biomedical Engineering 2018; Springer: Singapore, 2019; pp. 183–189.
- 123. Seetha, J.; Raja, S.S. Brain tumor classification using convolutional neural networks. *Biomed. Pharmacol. J.* **2018**, *11*, 1457. [CrossRef]
- 124. Kaldera, H.; Gunasekara, S.R.; Dissanayake, M.B. Brain tumor classification and segmentation using faster R-CNN. In Proceedings of the 2019 Advances in Science and Engineering Technology International Conferences (ASET), Dubai, United Arab Emirates, 26 March–10 April 2019; pp. 1–6.
- 125. Sarhan, A.M. Brain tumor classification in magnetic resonance images using deep learning and wavelet transform. *J. Biomed. Sci. Eng.* **2020**, *13*, 102. [CrossRef]
- 126. Kalaiselvi, T.; Padmapriya, S.; Sriramakrishnan, P.; Somasundaram, K. Deriving tumor detection models using convolutional neural networks from MRI of human brain scans. *Int. J. Inf. Technol.* **2020**, *12*, 403–408. [CrossRef]
- 127. Rajagopalan, N.; Narasimhan, V.; Vinjimoor, S.K.; Aiyer, J. Deep CNN framework for retinal disease diagnosis using optical coherence tomography images. *J. Ambient. Intell. Humaniz. Comput.* **2021**, *12*, 7569–7580. [CrossRef]
- Elnour, M.; Himeur, Y.; Fadli, F.; Mohammedsherif, H.; Meskin, N.; Ahmad, A.M.; Petri, I.; Rezgui, Y.; Hodorog, A. Neural network-based model predictive control system for optimizing building automation and management systems of sports facilities. *Appl. Energy* 2022, 318, 119153. [CrossRef]
- 129. Park, J.; Lee, J.; Sim, D. Low-complexity CNN with 1D and 2D filters for super-resolution. *J. Real-Time Image Process.* 2020, 17, 2065–2076. [CrossRef]
- Maharana, K.; Mondal, S.; Nemade, B. A review: Data pre-processing and data augmentation techniques. *Glob. Transitions Proc.* 2022, 3, 91–99. [CrossRef]
- Blind, K.; von Laer, M. Paving the path: Drivers of standardization participation at ISO. J. Technol. Transf. 2022, 47, 1115–1134. [CrossRef]
- 132. Stergiopoulos, V.; Vassilakopoulos, M.; Tousidou, E.; Corral, A. An Application of ANN Hyper-Parameters Tuning in the Field of Recommender Systems; Technical Report; Data Structuring & Engineering Laboratory, University of Thessaly: Volos, Greece, 2022.
- 133. Shen, A.; Han, X.; Cohn, T.; Baldwin, T.; Frermann, L. Optimising equal opportunity fairness in model training. *arXiv* 2022, arXiv:2205.02393.
- 134. Gulzar, Y. Fruit image classification model based on MobileNetV2 with deep transfer learning technique. *Sustainability* **2023**, 15, 1906. [CrossRef]
- 135. Zhang, G.; Xu, S.; Li, J.; Guo, A.J. Group-based network pruning via nonlinear relationship between convolution filters. *Appl. Intell.* **2022**, *52*, 9274–9288. [CrossRef]
- Okey, O.D.; Melgarejo, D.C.; Saadi, M.; Rosa, R.L.; Kleinschmidt, J.H.; Rodríguez, D.Z. Transfer learning approach to IDS on cloud IoT devices using optimized CNN. *IEEE Access* 2023, 11, 1023–1038. [CrossRef]
- 137. Costa, V.G.; Pedreira, C.E. Recent advances in decision trees: An updated survey. Artif. Intell. Rev. 2023, 56, 4765–4800. [CrossRef]
- 138. Zhang, H.; Jiang, L. Fine tuning attribute weighted naive Bayes. *Neurocomputing* 2022, 488, 402–411. [CrossRef]
- 139. Bansal, M.; Goyal, A.; Choudhary, A. A comparative analysis of K-nearest neighbor, genetic, support vector machine, decision tree, and long short term memory algorithms in machine learning. *Decis. Anal. J.* **2022**, *3*, 100071. [CrossRef]
- 140. Balyan, A.K.; Ahuja, S.; Lilhore, U.K.; Sharma, S.K.; Manoharan, P.; Algarni, A.D.; Elmannai, H.; Raahemifar, K. A hybrid intrusion detection model using ega-pso and improved random forest method. *Sensors* **2022**, *22*, 5986. [CrossRef]
- 141. Ikotun, A.M.; Ezugwu, A.E.; Abualigah, L.; Abuhaija, B.; Heming, J. K-means clustering algorithms: A comprehensive review, variants analysis, and advances in the era of big data. *Inf. Sci.* 2022, 622, 178–210. [CrossRef]
- Khalil, R.A.; Saeed, N.; Masood, M.; Fard, Y.M.; Alouini, M.S.; Al-Naffouri, T.Y. Deep Learning in the Industrial Internet of Things: Potentials, Challenges and Emerging Applications. *IEEE Internet Things J.* 2021, *8*, 11016–11040. [CrossRef]
- 143. Deng, G.; Duan, X.; Tang, M.; Zhang, Y.; Huang, Y. Non-interactive and privacy-preserving neural network learning using functional encryption. *Future Gener. Comput. Syst.* **2023**, 145, 454–465. [CrossRef]
- 144. Rubab, S.; Khan, M.M.; Uddin, F.; Abbas Bangash, Y.; Taqvi, S.A.A. A Study on AI-based Waste Management Strategies for the COVID-19 Pandemic. *ChemBioEng Rev.* 2022, *9*, 212–226. [CrossRef]
- 145. Malaysha, S.; Moreb, M.; Zolait, A. Detecting Network Traffic-based Attacks Using ANNs. Int. J. Comput. Digit. Syst. 2023, 13, 131–137. [CrossRef]
- 146. Arsalan, M.; Di Matteo, D.; Imtiaz, S.; Abbas, Z.; Vlassov, V.; Issakov, V. Energy-Efficient Privacy-Preserving Time-Series Forecasting on User Health Data Streams. In Proceedings of the 2022 IEEE International Conference on Trust, Security and Privacy in Computing and Communications (TrustCom), Wuhan, China, 9–11 December 2022; pp. 541–546.
- Al-Shqeerat, K.H.A. Securing a Question-Based Multi-factor Authentication System Using LSB Steganography Technique. In International Conference on Business and Technology; Springer: Cham, Switzerland, 2022; pp. 1118–1128.
- 148. Kalam, S.; Yousuf, U.; Abu-Khamsin, S.A.; Waheed, U.B.; Khan, R.A. An ANN model to predict oil recovery from a 5-spot waterflood of a heterogeneous reservoir. *J. Pet. Sci. Eng.* **2022**, *210*, 110012. [CrossRef]

- 149. Krichen, M.; Ammi, M.; Mihoub, A.; Almutiq, M. Blockchain for modern applications: A survey. Sensors 2022, 22, 5274. [CrossRef]
- 150. Bharany, S.; Sharma, S.; Khalaf, O.I.; Abdulsahib, G.M.; Al Humaimeedy, A.S.; Aldhyani, T.H.; Maashi, M.; Alkahtani, H. A systematic survey on energy-efficient techniques in sustainable cloud computing. *Sustainability* **2022**, *14*, 6256. [CrossRef]
- Al-juboori, A.M.; Alsaeedi, A.H.; Nuiaa, R.R.; Alyasseri, Z.A.A.; Sani, N.S.; Hadi, S.M.; Mohammed, H.J.; Musawi, B.A.; Amin, M.M. A hybrid cracked tiers detection system based on adaptive correlation features selection and deep belief neural networks. *Symmetry* 2023, 15, 358. [CrossRef]
- 152. Javaid, S.; Wu, Z.; Hamid, Z.; Zeadally, S.; Fahim, H. Temperature-aware routing protocol for Intrabody Nanonetworks. J. Netw. Comput. Appl. 2021, 183, 103057. [CrossRef]
- 153. Saeed, N.; Loukil, M.H.; Sarieddeen, H.; Al-Naffouri, T.Y.; Alouini, M.S. Body-Centric Terahertz Networks: Prospects and Challenges. *IEEE Trans. Mol. Biol. Multi-Scale Commun.* **2022**, *8*, 138–157. [CrossRef]
- 154. Fahim, H.; Javaid, S.; Li, W.; Mabrouk, I.B.; Al Hasan, M.; Rasheed, M.B.B. An efficient routing scheme for intrabody nanonetworks using artificial bee colony algorithm. *IEEE Access* 2020, *8*, 98946–98957. [CrossRef]
- 155. Javaid, S.; Wu, Z.; Fahim, H.; Mabrouk, I.B.; Al-Hasan, M.; Rasheed, M.B. Feedforward Neural Network-Based Data Aggregation Scheme for Intrabody Area Nanonetworks. *IEEE Syst. J.* 2020, *16*, 1796–1807. [CrossRef]
- 156. Fahim, H.; Li, W.; Javaid, S.; Sadiq Fareed, M.M.; Ahmed, G.; Khattak, M.K. Fuzzy logic and bio-inspired firefly algorithm based routing scheme in intrabody nanonetworks. *Sensors* **2019**, *19*, 5526. [CrossRef] [PubMed]
- 157. Cui, F.; Yue, Y.; Zhang, Y.; Zhang, Z.; Zhou, H.S. Advancing biosensors with machine learning. *ACS Sens.* **2020**, *5*, 3346–3364. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.